

Reviewer #1:

Scientific Quality: Grade B (Very good)

Language Quality: Grade B (Minor language polishing)

Conclusion: Accept (General priority)

Specific Comments to Authors: Thank you for the opportunity reviewing this interesting case report. Since the case is in its form unique, it should be published in the WJG. There is only one formal point: beside the images of the histopathologic analyses, I would recommend to summarize the major findings of the pathologist in a table. This would give a better overview about the course of the disease.

**Reply:** I have summarized the Pathologic findings in Table1.

Science editor: Issues raised:

- (1) The “Author Contributions” section is missing. Please provide the author contributions;
- (2) The authors did not provide the approved grant application form(s). Please upload the approved grant application form(s) or funding agency copy of any approval document(s);
- (3) The authors did not provide original pictures. Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor;
- (4) PMID and DOI numbers are missing in the reference list. Please provide the PubMed numbers and DOI citation numbers to the reference list and list all authors of the references. Please revise throughout.

**Reply:** I have added “Author Contributions” section in the manuscript. I have uploaded the approved grant application form. I have provided the original pictures in PowerPoint. I have listed all the authors and provided the PubMed numbers and DOI citation numbers in the references. Thank you very much for your review.